Author's response to reviews

Title: The development of effective behaviour change interventions to support the use of malaria rapid diagnostic tests by Tanzanian clinicians

Authors:

Clare I.R. Chandler (clare.chandler@lshtm.ac.uk)  
Judith Meta (judmeta@gmail.com)  
Celia Ponzo (celia.ponzo@gmail.com)  
Fortunata Nasuwa (mankaray@yahoo.com)  
John kessy (johnkessy@gmail.com)  
Hilda Mbakilwa (tufuwage@hotmail.com)  
Ane Haaland (ane.haaland@gmail.com)  
Hugh Reyburn (hugh.reyburn@lshtm.ac.uk)

Version: 4  
Date: 6 May 2014

Author's response to reviews:

n/a